The Moderating Effect of *zakāh* Distribution on the Economic Well-being of the Poor: An Analysis in Kelantan, Malaysia

Ahmad Fahme Mohd Ali^a, W.A. Amir Zal^b, Nurhanan Ab. Hamid^c, Tengku Fauzan Tengku Anuar^d, Hafizi Mat Salleh^e & Siti Asma Md. Rasdi^e

^aSenior Lecturer, Faculty of Hospitality, Tourism and Wellness ^bAssociate Professor, Centre for Language Studies and Generic Development ^cSenior Lecturer, Faculty of Agro Based Industry ^dSenior Lecturer, Faculty of Creative Technology and Heritage ^eInstitute for Poverty Research and Management University Malaysia Kelantan, Malaysia.

Received: 06 September 2020, Revised: 10 March 2021; 19 June 2021, Accepted: 14 July 2021

ABSTRACT. The main objective of this paper is to study the moderating effect of $zak\bar{a}h$ distribution on the family structure and the economic well-being of the poor in the state of Kelantan in Malaysia. The state of Kelantan was selected for the survey because it has the highest incidence of poverty in Peninsular Malaysia. The analysis of this study is based on Variance Based Structural Equation Modelling (VB-SEM) technique, while the data mining process, descriptive statistics and demographic data of this study were conducted using SPSS software. This study revealed that $zak\bar{a}h$ assistance and family structure dimensions positively and significantly affect economic well-being. The result also indicates that the direct effect of age, marital status, and gender is not significant due to the insufficient economic well-being that does not meet their basic needs. The mediating role of $zak\bar{a}h$ indicates that age, marital status, and gender have both significant indirect effect of family structure on economic well-being is fully mediated by $zak\bar{a}h$.

KEYWORDS: Structure Equation Model (SEM), Poor and Needy, *zakāh* Distribution, Poverty, Kelantan, Malaysia.

JEL CLASSIFICATION: D13, D31, I32 KAUJIE CLASSIFICATION: E12, E15, N4

1. Introduction

One of the fundamental dimensions of human development is economic well-being (OECD, 2013, p. 27). The basic need of all human beings is the pursuit to fulfil their basic needs, such as food and shelter. Copestake (2007) advocated that well-being is a state of society in which people's basic needs are met, they can act effectively and meaningfully in pursuit of their goals, and they would feel satisfied with their lives (Copestake, 2007, p.18). A high level of wellbeing means that the individual or group leads a positive life, while a low level of well-being is associated with adverse events (Guttman & Levy, 1982, p. 163). Although the concept of well-being is widely used, there is no commonly agreed definition. However, the OECD states that the key components of people's well-being are their income and expenditure (OECD, 2013, p.27).

 $Zak\bar{a}h$ is one aspect of Islamic economy that has direct substantial implications for the Muslim. The primary purpose of *zakāh* is to redistribute society's wealth among the poor (Qardhawi, 2000, p.6). The method requires Muslims (well-to-do) to distribute a small portion of their wealth to eight types of people $(zak\bar{a}h \text{ recipients})^1$ to reduce poverty and achieve their economic emancipation (Ahmed, 2004, p.15). In Islam, poverty is considered to have been eradicated once the bare minimum standard of living for everyone has been achieved (Qardhawi, 2000, p. 264). This concept would be absolute if it were viewed in terms of a fixed quantity of basic material needs to be met. However, Islam does not define a fixed criterion of poverty that applies to everyone, so the concept of poverty in Islam is broader and covers more than food and services (Qardhawi, 2000, p.14).

This study focuses on analyzing the moderating impact of $zak\bar{a}h$ assistance on the economic wellbeing of the poor as this aspect is crucial for a longterm economic policy. Malaysia has been facing a burden on the family structure due to the aging of society, high urban-rural cost of the living gap, gender inequality, high number of single parents, and increase in the number of nuclear families, which are likely to have a significant impact on public revenues and expenditures. Due to tight budgetary constraints, non-growth enhancing expenditures may crowd out outlays that could stimulate economic growth and would undoubtedly cause "budget crowding out" (Roloff, 2002, p.150). Therefore, the question of which government spending can promote permanent movements in economic growth rates becomes increasingly important. More specifically, this study has two objectives. First, it attempts to measure the effect of zakāh on improving the economic wellbeing of the poor. Second, this study attempts to examine the effect of family structure in reducing the burden on the economic well-being of the poor among zakāh recipients.

The structure of this paper is composed of the present section which deals with the introduction of the study. Section II discusses the method of $zak\bar{a}h$ distribution in Kelantan, which is the study area of this study. Section III explains the concept of economic well-being among the poor in Malaysia. Section IV presents the literature review and the development of the hypothesis. Section V reviews the literature and the development of the hypothesis. Section VI discusses the research design for this study. Section VII presents the empirical results and discussion of this study, and Section VI concludes the paper and suggests future studies and policy recommendations.

2. Zakāh Distribution in Kelantan

Zakāh assistance for the poor and needy in Kelantan can be classified into two categories: long-term and short-term assistance. Long-term assistance includes training and skill enhancement programs, education aid, business aid and housing aid, which are given on a one-time basis. Short-term assistance includes monthly aids such as *Eidul-Fitr* and *Eidul Adha*, and emergency aids that can be one-time or continuous (monthly). Among the highest zakāh assistance are the monthly *zakāh* aids paid to the poor and needy whose amount is based on *Haddul Kifayah* (*Zakāh* Poverty Line). *Zakāh* Poverty Line (ZPLI) determines the level of need required by the household based on their daily needs. It is calculated based on various variables such as the number of members in a

⁽¹⁾ Allah s.w.t says in the Qur'an, which means: "Zakāh is for: the poor, the destitute, those who collect it, reconciling people's hearts (new converts), frees slaves, spending in the way of Allah, and travelers. It is a legal obligation from Allah. Allah is all knowing, all wise" [Qur'an,9: 60].

household, and their age groups, etc. The Malaysian government, through *JAWHAR*, has defined the five main elements to determine the ZPLI (necessity) of a household, based on *Maqasid al Sharia* (human needs). Establishing the ZPLI will facilitate the *zakāh*

application process, as the committee will be able to immediately identify the applicants' situation, i.e., whether they are poor or very poor. Table 1 shows the amount of ZPLI in Kelantan.

Table (1). Kelantan Zakāh Pov	erty Line (<i>Had Kifayah</i>)) Per Month (2019) (MYR).
-------------------------------	----------------------------------	---------------------------

Category	Urban	Rural
Family Head	¹ 297.00/	¹ 182.00/
	² 549.00	² 380.00
Adults (working)	236.50	146.00
Adult (not working / not in school)	139.50	87.50
Adult (in school)	274.00	203.50
Adolescents (in school)	236.50	166.50
Children (in school)	179.50	130.50
Children (not in school)	103.00	77.50
Total	¹ 1,169.00/ ² 1,718.50	¹ 811.50/ ² 1191.50

¹Single family house; ²Rent house.

Table 1 above shows that the determination of the zakāh poverty line is based on the necessity of a household in Kelantan. For example, a family with both parents working, with an employed adult above 18 years old, a teenager aged 16 and a child aged 6 and live in paid house in urban area, suggests that the necessity of this household is MYR1361.50. If the monthly household income is RM1,500, then this family does not qualify for zakāh distribution because the household income is higher than the ZPLI (higher than MYR1,361.50). Nevertheless, if the household income is MYR 1,000, it qualifies for zakāh distribution. The Kelantan Zakāh Centre (MAIK) will distribute the shortfall (ZPLI gap) of MYR 361.50 to this family to fulfil their basic needs. In addition, if there is a situation such as a disabled or chronically ill household member, the total amount of ZPLI increases.

The Kelantan State Islamic Religious Department (MAIK) periodically reviews the ZPLI based on data from the National Poverty Line Index (PLI). At present, the Kelantan $Zak\bar{a}h$ Centre uses the Cost of Basic Needs (CBN) method to estimate the $zak\bar{a}h$ poverty line. The CBN method involves specifying a set of foods to be consumed based on the minimum caloric needs of humans. The cost of the total food consumption is then added to the non-food expenditures of poor households to reach the total poverty

Source: MAIK, 2020.

line that satisfies basic human needs. However, from Table 1, it is evident that higher $zak\bar{a}h$ distributions are for non-food items in both urban and rural areas. Although previous studies reveal that most rural areas' expenditures are higher on food items than urban areas, the analysis of the same pattern between food and non-food expenditures in both areas will create a biased poverty line. Therefore, it is important to analyse the pattern of food and non-food expenditures arong this group; ignoring the importance of food and non-food expenditure can make poverty alleviation (i.e. ZPLI) in Kelantan ineffective.

3. Economic well-being among the Poor and Needy in Malaysia

Malaysia has enjoyed a long period of economic growth since independence followed by a declining poverty rate. Cyclical downturns occurred during the Asian crisis of 1997/98 and again in 2017 but it was less severe that year. The current GDP growth in 2017 has returned to double-digit levels (World Bank, 2018, p. 4). This indicates a higher standard of living that shows significant growth in the public's overall level of income and consumption. In addition to the increase in the magnitude of income and expenditure may create some changes in the composition and types of goods and services demanded. Social and demographic changes have implications for the behavioural patterns of the Malaysian population (Ishida et al., 2003, p. 66). For example, in 2014, the average monthly household income and expenditure were MYR 6,141 and MYR 3,578, respectively. The average monthly household income increased to MYR 6,958, or 6.2% growth per annum, while average monthly household expenditure increased to MYR 4,033, or 6% per annum in nominal value in 2016 (DOS, 2018, p.15). During this period, house-holds in urban areas recorded an increase in median income (6.6%) and expenditure (5.8%) per annum, while rural households also increased by 6.5% (income) and 5.7% (expenditure) per annum (DOS, 2018, p. 15).

Most studies on the economic well-being focus on a single dimension of household economic wellbeing (Ada Ferrer-i-Carbonell, 2004, p.645; De Neubourg, De Milliano et al., 2014, p.8; Whelan, Nolan et al. 2014, p.188). However, the concept of economic well-being can be better understood by looking simultaneously (in a multidimensional way) at household income and consumption expenditure. Although there are some problems with the definition and measurement scope, all the essential elements can be included, and at the same time, the consistency of the various measures can be maintained by looking at them comprehensively. Meanwhile, measuring poverty has evolved over time from using income and expenditure-based methods to other methods that are not easily monetized or measured. These changes have been driven by evolving definitions of poverty and the new phenomenon of poverty that is included. A good example of the changing definition of poverty is the advent of multidimensional poverty measurement (Alkire & Foster, 2013, p.31; Alkire, 2009, p.4). According to Selamah Adbullah (2010). changes in poverty definitions are due to social and demographic changes that affect spending and consumption behaviour patterns (Selamah Abdullah Yusof & Jarita Duasa, 2010, p. 91). A clear understanding of how Malaysians can acquire and spend their income should play a vital role in the government's implementation of effective strategies to achieve the country's economic goals.

4. Empirical Literature Review and Hypothesis Development

4.1. Impact of age on the economic well-being of the poor

The development process in Malaysia has brought about a significant socio-economic and demographic transformation. The decline in fertility and mortality has resulted in an increase in the population's life span. As a result, the number of elderly people is increasing. The ageing population, which affects Malaysia's demographic profile, has a significant impact on its economic well-being. The main challenge faced by older Malaysians is the lack of savings upon retirement. Some public policies are insufficient to provide benefits to the elderly and retired, such as the government's welfare programs that need to be correctly targeted and paid out effectively (Abd Samad, S., & Mansor, N., 2017, p. 140). At the same time, Mohd. S, (2014) found that the probability of the elderly living in poverty is relatively low at about 2.6%. He, however, suggests that the probability of living in poverty changes with different socio-demographic characteristics (Mohd. S, 2014, p. 2). Therefore, age classification is vital for policymakers to investigate the poverty level among the society, including their early age, productivity, or retirement. The age classification in this study is based on the Malaysian food requirements category obtained from the Ministry of Health Malaysia (MOH, 2005). This study classified age into three subgroups, which means different income generation and expenditure levels. The first category (19-24 years old) comprises people who have just finished secondary school and are very young to work and start a family. The second category (25-59) comprises people who are of working age, most of whom have skills and work experience. The final category (60+) is those of retirement age.

H1: Age has a positive effect on the economic well-being of the poor.

4.2 Impact of family size on the economic well-being of the poor

Family size indicates a strong negative correlation between household size and economic well-being (Lanjouw, P. and Ravallion, M., 1995, p.1). The issue of poverty estimates as a function of household size behaviour has recently attracted considerable attention (Buhmann, B. et al. (1988, p.115), Coulter et al. (1992, p.1068), Dreze and Srinivasan (1997, p.218), Lancaster, et al. (1999, p.178). Most of these studies have focused on economies of scale on household size. Lanjouw and Ravallion's (1995, p.12) study in Pakistan shows that the presence of economies of size in consumption is likely to counteract the widely held view that large families tend to be poorer in developing countries. Mok (2009, p.94) studied economies of scale among the poor in Malaysia and found that the poor, who typically have larger households, choose to allocate resources to private consumption such as education, transport and medicine due to the large savings in public consumption.

H2: Family size has a positive effect on the economic well-being of the poor.

4.3 Impact of the region on the economic well-being of the poor

Previous studies on poverty in Malaysia demonstrate that poverty is more severe among rural households and those who work in the agricultural sector (Mok, 2009, p.47; Geda et al., 2005, p.11). Mok et al. (2009, p.47) shows that poverty in Malaysia is higher in rural areas than in urban areas. Mukaramah et al., (2011, p.21) studied urban and rural public expenditure in Malaysia and found that education expenditure has a very significant impact on increasing the household income, mainly in urban Malaysian households. The impact of public expenditure in education on urban public investment in education, personal needs, health, and general administration are also favourable programs not only for urban households but also for rural.

H3: The region has a positive effect on the economic well-being of the poor.

4.4 Impact of marital status on the economic wellbeing of the poor

The marital status of the household head in this study had created a higher intention in the 21st century which is not as popular as in the previous century (Mergenhagen et al., 1985, p.53). This connection may be derived from certain factors that are associated with marriage. Dunga, S. H. (2017, p.28) found out that married households have higher incomes than single, divorced, and widowed households in South Africa. In addition, a married family can significantly improve the family income and wealth (Waite LJ., 1995, p. 505; Light A., 2004, p.505; Zagorsky JL, 2005, p.420). They can also share resources while reducing their economic hardship (Bauman K., 1997, p.320). While being a single parent will have negative economic consequences which are mainly borne by women (Zagorsky J.L., 2005, p.420; Holden KC, 1991, p.70), the family can provide social support and other noneconomic resources that help them in times of economic shock or uncertainty (Cohen S, Syme L, 1995, p.680; Hobfall S. E., et. al., 1996, p.414), although the benefits may be less for women than men (Cutrona C. E., 1996, p.47; Neff L. A. et. al., 2005, p.80).

H4: Marital status has a positive impact on the economic well-being of the poor.

4.5 The impact of gender on the economic wellbeing of the poor

Over the last three decades, many women's advocates have argued that women tend to be poorer than men, which stems from the concept of the "feminization of poverty". This idea has become popular both in poverty analyses and poverty alleviation strategies. Meanwhile, earlier evidence shows that femaleheaded households enjoy less social, economic, and demographic benefits than male-headed counterparts, making the former poorer than the latter (Barros et al., 1997, p.24). In addition, Fuwa (2000, p.22) shows that single mothers had a higher deprivation in both income and non-income dimensions than maleheaded households. Pahl (1995) found that when men are typically in charge contribute more to domestic finances, women consequently have lower personal expenses than their male partners (Pahl J., 1995, p.370).

H5: Gender has a positive impact on the economic well-being of the poor.

4.6 Moderating effect of zakāh distribution on the economic well-being of the poor family's structure

Sohag, K., (2015, p.14) shows that $zak\bar{a}h$ distribution in Bangladesh manages to increase the purchasing capacity of rural recipients. The results also show that household income had a significant and positive effect on total household expenditure. It was assumed that recipients' saving capacity would increase by taking part in the $zak\bar{a}h$ program. The study confirmed that training of $zak\bar{a}h$ recipients was positively and significantly related to the total household expenditure. The study also confirmed that household income was positively and significantly related to savings. This indicates that household savings increase with income. In fact, $Zak\bar{a}h$ distributed to the poor and needy is an additional income for this group of people, who are given the extra purchasing power directly.

- H6: *Zakāh* distribution reinforces the positive effect of age on the economic well-being of the poor.
- H7: *Zakāh* distribution reinforces the positive effect of family size on the economic well-being of the poor.
- H8: *Zakāh* distribution reinforces the positive effect of the region on the economic well-being of the poor.
- H9: *Zakāh* distribution reinforces the positive effect of marital status on the economic well-being of the poor.
- H10: Zakāh distribution reinforces the positive effect of gender on the economic well-being of the poor.
- H11: Zakāh distribution has a positive effect on the economic well-being of the poor.

Figure 3 illustrates this study's conceptual model, which highlights the moderating effect of zakāh distribution on their economic well-being. Many studies have examined the effect of public expenditure (i.e., $zak\bar{a}h$) on improving economic wellbeing and reducing poverty among the poor, whether it is a multidimensional or unidimensional construct (Alkire Sabine et al., 2015, p. 216; Aaberge, R., et al., 2015, p. 26). This discrepancy is due to the inconsistency of perceptions regarding the phenomenon of the economic well-being of the poor (Xu, D, et al., 2017, p.67; Cheung, F., & Lucas, R. E., 2016, p.337). However, regarding the growing controversy within the fundamental concept of economic well-being, and in line with the aims and objectives of this study on the concept of economic well-being as a unidimensional construct, this means that all the separate dimensions within the economic well-being construct (income and expenditures) are integrated and analyzed to reflect a single variable: economic well-being.





5. Research Method

5.1. Measurement

This study uses the 5-point Likert scale questionnaire in Bahasa Melayu. The scales range from disagree (0) to agree (5). To study the effectiveness of $zak\bar{a}h$. five items were adapted from Fuadah et al. (2015, p. 46) and Fahme et al. (2015, p. 361). The reliability of the items in this measurement scale ranges from 0.88 to 0.89 (Fahme et al., 2015, p. 359). Examples of these items include sufficiency on monthly zakāh, type of zakāh, the period of zakāh assistance received, and the effect of zakāh distribution on family income and expenses. The 25 items were adapted from Manning et al. (2003, p. 10), Andrew Morrison et al. (2007, p. 200), Banovcinova, Andrea (2014, p. 150) and Fahme et al. (2018, p. 53) to measure the dimensions of family structure. The reliability of this measurement scale ranges from 0.60 to 0.83. Examples of these items include the effect of family structure (age, family size, gender, region, and marital status) on economic activity, the impact of family members on the family economy, the contribution and effect of other siblings or relatives on the family economy, and the effect of family structure on family's wages and economic capacity. Five separate items were adapted from Bauman, K. (2000, p. 5) and Mariano Rojas (2004, p. 5) to measure the dimension of economic well-being; these items include, for example, the effect and adequacy of income on expenditure, the effect of other assistance received (besides $zak\bar{a}h$), the sufficiency of monthly food and non-food expenditure and the possession of durable goods. The reliability of this measurement scale is between 0.78 and 0.88.

This study measures the economic well-being of the poor and needy based on the amount of income they receive and the expenditures they make. The sources of income are divided into four, namely income from wages and salaries, transfers and contributions from others (such as relatives), income from property and income from any economic activity. To obtain the amount of total household income, all types of income of all household members are transformed into monetary values. The expenditure amount is obtained from two sources which are 1) food expenditures are classified into two categories: (1) food purchases, including food and (2) food consumed outside the home. For nonfood expenditures, nine types of non-food expenditures are collected, namely (1) housing, including utilities and housing contents and services; (2) clothing and footwear; (3) medical care; (4) transportation; (5) education; (6) religion; (7) miscellaneous goods and services, including recreation and insurance; and (8) other expenditures, including other payments, savings, fines, and the money given to others. The items in this section were adapted from the Malaysia Statistics Household Expenditure Survey (2015).

5.2. Sample size and data collection method

The state of Kelantan was selected as the study area for this study. This state has the highest incidence of poverty in Peninsular Malaysia (UPR, 2017, p. 4). In addition, this state has many resources to address poverty, such as zakah, which is one of the highest zakāh collection states in Malaysia and can be used as a poverty alleviation fund (MAIK, 2016, p. 5). However, to implement a poverty alleviation policy, policymakers must know the consumption pattern at the household and individual level, especially among the poor (Mok, T. et al., 2007, p. 190). The respondents of this study were selected from the 2016 zakāh recipients list of the Kelantan Islamic Religious Council (MAIK) in the category of poor and needy, which is the lowest income quartile in Kelantan. The primary criteria for selecting respondents are those receiving zakāh assistance (long-term or short-term) from MAIK for more than one year. Thanks to the cooperation and assistance of the MAIK zakāh office, the survey covered all ten districts of Kelantan. The multi-stage stratification method was selected and applied because there are three stages in the process of selecting respondents, namely district (ten districts), region (urban-rural) and gender of the household head (male-female). This method is appropriate because it requires the total population to be divided into strata or sub-population, after which the samples are selected randomly but independently (Etikan, I., & Babtope, O., 2019, p.051). The sampling frame was stratified by region and gender of the household head. The list of zakāh recipients was collected from Kelantan Zakah Department (MAIK), and it will be used as a location reference for the respondents' information.

In order to ensure that the items in the questionnaire achieve the aims, scope and objectives of this study, three senior researchers and experts from the Faculty of Economics and Muamalat, University Sains Islam Malaysia (USIM) and various zakah centres were invited to review, evaluate and improve the questions. In addition, a pilot study was conducted in Kota Bharu, Bachok and Pasir Putih to test the validity and reliability of the questionnaire and ensure that it can be understood and answered by the respondents while estimating the time required for data collection procedures. Fifty zakāh recipients were involved in the pilot study, which is consistent with existing literature highlighting the relevance of 50 respondents in a pilot test (Ogbeibu, S., 2018, p.338). The pilot study results indicated that out of 40 items, 6 items loaded less than the recommended cutoff (0.70) after analysis using SPSS software. Therefore, they were removed, reducing the total number of indicator items to 34 (Lee, C., 2018, p.303; Sarstedt et al., 2014, p.157; Yong & Pearce, 2013, p.80).

Nevertheless, prior to estimation, all observations of each variable were weighted using as a sample weight variable the rate of masculinity within each household, compared to the gender rate of the regional population (population data from MAIK recipients list, 2016). This operation assigns a load to all the values in the data set to make the sample more representative of the entire regional population, leading to a much more reliable and accurate result (Hahs-Vaughn & Lomax, 2006, p.185; Ogbeibu, S., 2018, p.339; Bollen, K. A., 2013, p.1237). In addition, a stratification variable based on the district of origin of each household creates mutually exclusive and internally homogeneous subgroups that allow for a lower standard error in the estimation. Both stratifycation and weighting are applied through the survey data option available in the statistical package. The sample size determinant method of Krejcie and Morgan (1970, p. 609) was used to guide the sample size determinant and obtain a stratified proportional sampling of zakāh recipients among the poor and needy. Of the 300 questionnaires that were distributed, 288 were returned, complete and suitable for analysis. This response rate of 96% is consistent with the existing literature (Mistry, R. S., 2002, p.938; Utsey et al., 2008, p.52). The respondents' profiles in this study are based on the household head's income, family size, region, gender, education level, age and marital status. The justifications for the respondents' profiles are explained in Table 1.

Profile	Category	Frequency	Percentage	Cumulative percentage
Pagion	Urban	158	55	55
Region	Rural	130	45	100
	19 - 24	82	28	28
Household Head Age	25 - 59	132	46	74
	60 years and over	74	26	100
	4 to 1	100	35	35
Family Size	5 to 8 years	88	31	65
	9 and up	100	35	100
	Not married	55	19	19
Household Head Marital Status	Married to	71	25	44
	Married but living apart	37	13	57
	Widow(er)	58	20	77
	Divorced	67	23	100
Household Hood Condon	Male	129	45	45
nousenoid nead Gender	Female	159	55	100
Total		288		100

Table (1). Respondent Profile in the Study.

Source: Research Question.

The respondents in this study were 55% (158) from urban areas and 45% (130) from rural areas (M=1.93, SD=0.78). The age of the respondents ranged from 19 to over 60 years, with females accounting for 55% and males 45%. The marital status of the respondents indicates the unmarried (19%), married (25%), married but separated (13%), widowed (20%) and divorced (23%).

5.3. Data analysis

Smart PLS 3.0 software was used for this study's exploratory and confirmatory factor analysis (EFA and CFA), where the Variance Based Structural Equation Modelling (VB-SEM) technique was applied. To analyze this study's data mining and descriptive statistics, SPSS software (version 2.2) was used; and to avoid the biases of ordinary methods, the respondents' identity remains completely anonymous to prevent them from changing their answers and make them less anxious (Podsakoff et al., 2003, p.880). The results of this study show that all VIF values are between 1.225 and 2.532 (< 3.3),

indicating that the standard method bias did not influence the participants' responses (Table 2).

6. Empirical Results and Discussion

The results of the descriptive statistics (Table 2) indicate that the mean of 2.95 out of 5 reflects that most of the respondents generally agree that zakāh distributions help and improve their economic wellbeing. The standard deviations are relatively close to each other, suggesting that the data are normally distributed and that the concepts were evenly distributed. The results in Table 2 show that Cronbach's Alpha (CA) (0.81 to 0.97) and Composite Reliability (CR) (0.82 to 0.94) exceeded the minimum requirement of 0.7, confirming the reliability and internal consistency of all constructs. The tolerance values were in the range of 0.395 to 0.816, which was well above the critical tolerance value (0.1) as suggested by Julie Pallant (2013, p.120). Therefore, it can be concluded that there was no serious multicollinearity between the variables used in this predictive model.

Table (2). Summary of Descriptive Statistics and Reliability and Validity of Measurement Model.

Constructs	N	Mean	Standard deviation	CA	CR	AVE	VIF	Tolerance
Age	288	2.96	1.591	0.837	0.876	0.567	1.225	0.816
Family Size	288	2.84	1.685	0.967	0.941	0.695	2.532	0.395
Region	288	2.67	1.021	0.939	0.938	0.683	2.112	0.473
Marital Status	288	2.72	1.318	0.916	0.910	0.594	2.014	0.497
Gender	288	2.53	1.541	0.872	0.903	0.557	1.771	0.565
Income	288	2.66	1.398	0.742	0.813	0.517	Endogenous	-
Expenditure	288	3.43	1.267	0.808	0.826	0.543	Endogenous	-
Zakāh Distribution	288	2.95	1.205	0.959	0.939	0.584	1.921	0.521

Note: Sample (N); CA (Cronbach's alpha); CR (composite reliability); AVE (average variance extracted); VIF (variance inflation factor).

The results of the initial output metrics that measure the external model measure between constructs and metrics (Figure 4) show that all metrics exceed the threshold of 0.7 (Sarstedt et al., 2014, p.156; Pallant. J, 2013, p.120; Field, A, 2013, p.706). This shows that all measurement items contribute critically to their individual constructs and that there is no confounding effect between the independent and dependent variables.



Figure (4). Measurement Model.

Heterotrait-monotrait ratio (HTMT) is used to test the discriminant validity in this study. The HTMT result (Table 3) shows that all constructs are less than one (< 1.0). All the constructs show a range of 0.621 to

0.849 (< 1.0). Thus, the result shows that all constructs are clearly independent of each other, and the criterion of discriminant validity has been met.

	AGE	SIZE	REG	MS	GEN	INC	EXP	ZKT
AGE								
SIZE	0.726							
REG	0.318	0.834						
MS	0.311	0.500	0.827					
GEN	0.404	0.365	0.546	0.771				
INC	0.421	0.285	0.232	0.406	0.849			
EXP	0.178	0.555	0.394	0.278	0.232	0.832		
ZKT	0.365	0.535	0.652	0.485	0.253	0.684	0.621	

Notes: Age (age of household head); Size (family size); REG (region) MS (marital status of household head); Gen (gender of household head); INC (income); POV (poverty); ZKT (zakāh).

The structural model in this study is measured based on empirical thresholds of statistical significance, effect sizes and R² values (Figure 5). The coefficient of determination (R²) in this study shows that a large degree of variance is explained in the economic wellbeing of the poor (0.716). Figure 5 also shows that *zakāh* (R²= 0.781) and region (R²= 0.775) have the highest degree of variance, while family size (R²= 0.636) has a moderate degree of variance and age (R^2 = 0.473), status (R^2 = 0.327) and gender (R^2 = 432) reflect a low degree of variance. Although there is some moderate and low degree of variance between constructs, the R^2 value indicates that it is statistically significant (t-statistic, 3.832; p≤0.01), suggesting that the six (6) exogenous constructs have a significant explanation for the variance in economic well-being (Hair et al., 2014, p. 113).

Figure (5). Structural Model and PLS-SEM Estimates.



The results in Figure 5 show that *zakāh* has the most significant positive impact and predictive ability on economic well-being, followed by region, family size, household age, gender and marital status of the household head. This corroborates the initial predictions of H1, H2, H3, H4 and H5 at p \leq 0.05 and p \leq 0.01, respectively. *Zakāh* has the highest significant positive effect on economic well-being. This had confirmed the initial postulation of H11 at p \leq 0.01.

CFA was conducted to measure the relationship between the observed variables and the underlying latent variables (Table 4). From Table 4, the CFA fit indices in this study show that all results exceeded the minimum cut-off value and meet the requirements of the acceptable level (P-value = 0.000, GFI = 0.884, CFI = 0.925, TLI = 0.889, RMSEA = 0.095). Thus, the CFA validity of this study indicates that it met the requirements of the acceptable level.

Table (4), CI II IIujustinent Index Results	Table	(4).	CFA	Adjustment	Index	Results
---	-------	------	-----	------------	-------	---------

P-value	GFI	FCI	TLI	RMSEA	ChiSq/df
0.000	0.884	0.925	0.889	0.095	1.621

Source: Research Question.

The results of the significance levels of the path coefficients (Table 5) show that age, family size and gender have significant positive effects on the economic well-being of the poor. This supports the initial predictions of H1, H2, H5 and H6 at p \leq 0.05 and p \leq 0.01, respectively. Furthermore, Table 5 also shows that *zakāh* assistance has a significant, positive and medium level moderating effect on the poor's economic well-being, which supports H11. Moreover, H3 and H4 are not supported since it is not statistically significant (p > 0.05) (p > 0.01).

Hypothesis	Constructs in the Structural Model	Effect Size (f ²)	T-Statistics	P-Values	Decision
H ₁ . Age of the household head affects the economic well-being of the poor.	Age <> Economic well-being	0.182	1.325	0.001	Supported
H ₂ . Family size affects the economic well-being of the poor.	Size <> Economic well-being	0.050	0.132	0.053	Supported
H ₃ . Region affects the economic well- being of the poor.	Region <> Econom- ic well-being	0.072	0.212	0.726	Not supported
H ₄ . Marital status affects the economic well-being of the poor.	Marital status <> Economic well-being	0.021	0.174	0.882	Not supported
H ₅ . Gender affects the economic well- being of the poor.	Gender <> Econom- ic well-being	0.198	2.531	0.001	Supported
H_{11} . Zakāh affects the economic well- being of the poor.	<i>Zakāh</i> <> Economic well-being	0.091	0.331	0.021	Supported

Table (5). Structural Model Path Analysis.

Source: Research Question

The moderating role of zakāh on the economic wellbeing of the poor in Kelantan indicates that the age of the household head has both a significant (indirect effect) and an insignificant (direct effect) effect on the economic well-being. The result shows that the indirect effects of age on economic well-being are fully mediated by zakāh (Table 6). The results corroborate that *zakāh* distribution did influence the economic well-being of zakāh recipients, which also supports the initial postulation of H6 at p<0.01. This conclusion was supported by the fact that the bootstrap interval (0.072; 0.136) of the indirect effect of age (0.109) on economic well-being through perceived zakāh did not contain the number zero. Moreover, the direct effect of the age of household head on economic well-being became insignificant (P=0.001) when associated with zakāh. According to this study, any effort to improve the economic wellbeing of the poor will be ineffective without considering the zakāh assistance.

 $Zak\bar{a}h$ partially mediated family size with a bootstrap interval of (0.084; 0.173), showing the indirect effect of family size (0.110) on economic well-being through $zak\bar{a}h$. Moreover, the direct effect

of family size on economic well-being was still significant (P < 0.001) when associated with *zakāh*. Thus, we can conclude that family size can still have a direct influence on the economic well-being even if it is not a recipient of zakāh. However, this effect (0.064) would be more significant when the family size of the poor is associated with $zak\bar{a}h$ (0.174). The effects of the region on economic well-being are partially mediated by zakāh, which means that the geographical condition of the family did affect its economic well-being. This was confirmed by a bootstrap interval (0.055; 0.091) of the indirect effect of region (0.107) on economic well-being through zakāh. Moreover, the direct effect of region on economic well-being is always significant (P < 0.001) when associated with the zakāh assistance. Therefore, it can be concluded that region could still have a direct influence on economic well-being without the zakāh assistance; however, this effect (0.085) would be smaller compared to when the region is associated with *zakāh* (0.107).

The marital status of the household head was fully mediated by $zak\bar{a}h$ with a bootstrap interval of (0.093; 0.182), showing a significant indirect effect

of the marital status of the household head (0.092) on economic well-being through $zak\bar{a}h$. Moreover, the direct effect of marital status on economic well-being becomes statistically insignificant (P=0.416) when combined with $zak\bar{a}h$. Meanwhile, a similar result occurs on the gender of the household head, where it was also fully mediated by $zak\bar{a}h$ with a bootstrap interval of (0.096; 0.187), which shows a significant indirect effect of the gender of the household head (0.112) on economic well-being through $zak\bar{a}h$. Furthermore, the direct effect of gender on economic well-being becomes statistically insignificant (P=0.351) when combined with $zak\bar{a}h$. In other words, the effect of the gender of the household head on his/her economic well-being did not occur directly, but had an impact on $zak\bar{a}h$, which in turn affected his/her or their economic well-being.

Hypothesis	Indirect effect		Direct effect		Bootstrap interval of the	Total	Type of
	Effect	<i>P-value</i> Effect <i>P-value</i>		P-value	indirect effect	effects	mediation
H ₆ : <i>Zakāh</i> mediates the effect of age on economic well-being.	0.109	<0.001 (significant)	0.041	0.535 (not significant)	(0.072; 0.136)	0.150	Total
H ₇ : <i>Zakāh</i> mediates the effect of family size on economic well-being.	0.110	<0.001 (significant)	0.064	<0.001 (sig- nificant)	(0.084 ; 0.173)	0.174	Partial
H ₈ : <i>Zakāh</i> mediates the effect of the region on economic well-being.	0.107	0.014 (significant)	0.085	<0.001 (sig- nificant)	(0.055; 0.091)	0.192	Partial
H ₉ : <i>Zakāh</i> mediates the effect of marital status on economic well- being.	0.092	0.021 (significant)	0.041	0.416 (not significant)	(0.093 ; 0.182)	0.133	Total
H ₁₀ : <i>Zakāh</i> mediates the effect of gender on economic well-being.	0.112	0.017 (significant)	0.054	0.351 (not significant)	(0.096 ; 0.187)	0.166	Total

Table (6). Moderator Effect Analysis.

Source: Research Question

7. Summary and Conclusion

7.1. Summary of results

The results of this study indicate some important findings that can improve $zak\bar{a}h$ distribution in Kelantan. $Zak\bar{a}h$ distribution shows that it has a partial effect (direct and indirect) on the well-being of the poor if it can be channelled through short and long term programs. The direct effect means that short-term programs such as monthly distribution of $zak\bar{a}h$ according to family size and region are effective and positively impact family welfare. The current method of $zak\bar{a}h$ distribution in Kelantan based on family size and region is found to have a positive effect. The total effect (indirect effect) results show that $zak\bar{a}h$ distribution has a positive effect on long-term programs, because $zak\bar{a}h$ distribution is not directly channelled to the poor, but through the

programs. For example, long-term programs such as vocational training programs organized by MAIK can improve the well-being of the poor by increasing their livelihoods, as they are able to work or start a business with their skills and knowledge. However, short-term $zak\bar{a}h$ assistance, such as monthly assistance, is not significant in improving their well-being because short-term $zak\bar{a}h$ assistance does not consider the age, marital status and gender of the recipient who has different economic well-being. Differences in needs and abilities according to age, marital status and gender of short-term recipients, which are not considered by direct distribution, make $zak\bar{a}h$ distribution less effective. Older people may receive a higher amount of $zak\bar{a}h$ than those of

productive age, while widows (female heads of household) and women may need to spend more than men because they have higher dependence from the children, especially after losing their partners.

This study shows that two likely causes contribute to the positive moderating effects of *zakāh*. First, the Kelantan Islamic Religious Council (MAIK), the body responsible for the management (collection and distribution) of zakāh in Kelantan, has made considerable efforts to manage zakāh, which can be verified by the impressive growth in zakāh collection and distribution from time to time. From 2014 to 2019, zakāh collection in Kelantan has increased from MYR 133 million (2014) to MYR 183 million (2018). while $zak\bar{a}h$ distribution also shows the impressive effort of MAIK where it has increased from MYR 151 million in 2014 to MYR 181 million in 2018 (JAWHAR, 2019, p. 12)⁽²⁾. Second, the implementtation of the zakāh poverty line (ZPL), had kifavah, in identifying the category of the poor and needy based on a minimum basic need, which takes into account all the basic needs of a family had made the zakāh distribution becomes more effective (Ali, A.F.M., 2014, p. 292). In contrast to the government poverty line income (PLI), the ZPL is more detailed in assessing the needs of the poor and needy as it is based on individual needs (i.e., age, gender and household size) while the former puts more emphasis on the total household poverty line. Thus, the implementation of ZPL can enable MAIK to determine the individual needs of the poor and needy in Kelantan (Fahme, 2015, p. 364).

Zak $\bar{a}h$ assistance based on the household head's age, marital status, and gender has an insignificant direct effect on economic well-being and has been totally mediated by zak $\bar{a}h$. This shows that zak $\bar{a}h$ distribution is only effective when it is relayed by other factors such as poverty reduction programs and skills enhancement courses that can produce a sustainable community support system in achieving human welfare. For example, the elderly may need zak $\bar{a}h$ assistance for their medicines, which can be provided by zak $\bar{a}h$ assistance. Widows or single parents and women heads of households may need zak $\bar{a}h$ assistance for skill enhancement programs that can ensure sustainable livelihoods. According to Ragayah (2007, p. 25), individuals have different food and non-food needs depending on their age and gender. She stresses that any poverty assessment should take into account the needs of individuals. In order to maintain the same nutritional standard, children need less food than adults, while women need less food than men, but at the same time, they may need to spend more on clothing. She suggested that a person with higher needs should have a higher poverty line than those with lower needs.

It is suggested that the method of direct distribution of zakāh should consider the age and gender of the household head and its members. If age, marital status and gender are not taken into account, the amount of zakāh given will be insufficient and will affect their economic well-being and, in the long run, also the process of livelihood acquisition, resulting in poverty being passed on to their generations. Furthermore, less effective distribution of zakāh may cause the public, mainly zakāh payers, to question the purpose and role of the zakāh institution in developing the Ummah's economy. If they have less faith and trust in the institution of zakāh to achieve positive results through its effective distribution, they may not pay their zakāh through zakāh institutions, which may slow down the process of developing the economic well-being of the Ummah due to the decrease in zakāh collection and poverty alleviation programs that can have a significant impact. Given the negative impact of marital status on the economic well-being, as shown in the findings of Pamela J. Smock et al. (1999, p. 800), it might be quite difficult for zakāh centers to directly channel the zakāh assistance and subsequently promote their economic well-being, while some degree of $zak\bar{a}h$ assistance is needed to get the single mother to work indepen-dently, which could then help her generate higher economic wellbeing. Although the different needs by age and gender suggest a different type of poverty line between these groups, there are significant direct effects based on region and family size, which result from a sufficient number of short and long term zakāh distributions. The importance of zakāh distribution can be seen as a higher positive moderating effect of zakāh that can improve economic well-being across all aspects of the family structure.

⁽²⁾ Department of Waqf, Zakāh and Hajj (2019). http:// intranet.jawhar.gov.my/spmj/public/zkt_statistik_map.php

References

- Aaberge, R. and Brandolini, A. (2015). Multidimensional Poverty and Inequality. Handbook of Income Distribution. 2: 141-216. Elsevier.
- Abd Samad, S. and Mansor, N. (2017). Population Aging and Social Protection in *Malaysia*. *Malaysian Journal* of Economic Studies, 50(2): 139-156.
- Ada Ferrer-i-Carbonell, (2004). "Income and Wellbeing: An Empirical Analysis of the Comparison Income Effect". *Journal of Public Economics*, 89 (2005): 997-1019.
- Ahmad Fahme Mohd Ali and Mohd Faisol Ibrahim (2018). Determination of Poverty among the Poor and Needy in a Developing Country. *Journal of Emerging Economies and Islamic Research*, 6(2): 47-61.
- Ahmad Fahme Mohd Ali, Muhammad Ridhwan Ab Aziz, (2014). "Zakāh Poverty Line Index and Gender Poverty in Malaysia: Some Issues and Practices", *International Journal of Business and Social Science*, 5(10): September 2014.
- Ahmad Fahme Mohd Ali, Zakariah Abd. Rashid, Fuadah Johari and Muhammad Ridhwan Ab. Aziz.(2015). The Effectiveness of Zakāh in Reducing Poverty Incidence: Analysis in Kelantan, *Malaysia*. *Asian Social Science*, **11**(21): 2015.
- Ahmed, H. (2004). The Role of Zakāh and Awqaf in Poverty Alleviation. Islamic Research and Training Institute. Occasional Paper no.8. Jeddah: Saudi Arabia.
- Al Qardawi, Y. (2000). *Fiqh al Zakāh* (Vol. 1). King Abdulaziz University.
- **Al Qardawi, Y.** (2000). *Fiqh al Zakāh* (Vol. **2**). King Abdulaziz University.
- Alkire, S. and Foster, J. (2009). An Axiomatic Approach to the Identification and Measurement of Multidimensional Poverty. Oxford Poverty & Human Development Initiative (OPHI).
- Alkire, S., Foster, J.E. (2013). A Report on the Measurement of Multidimensional Poverty in Mexico (No. 60). Oxford.
- Alkire, S., Roche, J.M., Ballon, P., Foster, J., Santos, M. E. and Seth, S. (2015). Multidimensional Poverty Measurement and Analysis. Oxford University Press, USA.
- Ali, M., Fahme, A., Noor, M., Aziz, A., Ridhwan, M., Ibrahim, M.F. and Johari, F. (2013). Impact of Zakāh Distribution on Poor and Needy Beneficiaries: An Analysis in Kelantan, Malaysia. *Australian Journal of Basic and Applied Science*, 7(13): 177-182, 2013.

- Al Jaffri Saad, R. and Haniffa, R. (2014), "Determinants of Zakāh Compliance Behavior", Journal of Islamic Accounting and Business Research, 5(2): 182-193.
- Arif, H.M., Alwi, K. and Tahir, A.M. (2011). Factors Influencing Companies towards Zakāh Payment: An Explanatory Studies. 2nd International Conference on Business and Economic Research (2nd ICBER 2011), Langkawi, Kedah.
- Ananat, Elizabeth and Michaels, Guy (2007). The Effect of Marital Breakup on the Income Distribution of Women with Children. *Working Paper*. Centre for Economic Policy Research, London, UK.
- Andrew Morrison, Dhushyanth Raju and Nistha Sinha. (2007). "Gender Equality, Poverty and Economic Growth". Policy Research Working Paper 4349. World Bank Gender and Development Group Poverty Reduction and Economic Management Network.
- Azam, M.; Iqbal, N. and Tayyab, M. (2014). "Zakāt and Economic Development: Micro and Macro Level Evidence from Pakistan", *Bulletin of Business and Economics*, 3(2): 85-95.
- Azura Mohd Noor, Ram Al Jaffri Saad (2016). "The Mediating Effect of Trust on the Relationship between Attitude and Perceived Service Quality towards Zakāh Compliance Behavior", *International Journal of Economics and Finance*, 2016, 6(S7): 27-31.
- Banovcinova, Andrea, Levicka, Jana and Veres, Martin. (2014). The Impact of Poverty on the Functioning of the Family System. Procedia - Social and Behavioral Sciences. 132. 148-153. 10.1016/j.sbspro.2014.04.291.
- Barker, D. K. (2014). Beyond Women and the Economy: Re-reading "Women's Work". Signs, 40(1).
- Barros, R., Fox, L. and Mendonca, R. (1997). "Female-Headed Households, Poverty, and the Well-being of Children in Urban Brazil", *Economic Development and Cultural Change*, 45(2): 231-57.
- Bauman K. (1997). Shifting Family Definitions: The Effect of Cohabitation and other Nonfamily Household Relationships on Poverty Measures. Poverty Research Institute *Working Paper* No. 1123-97. University of Wisconsin. Madison, WI; 1997.
- Bauman, K. (2000). The Effect of Work and Well-being on the Living Conditions of Single-Parent Households. Population Division, US Census Bureau.
- **Bollen, K.A., Tueller, S.J.** and **Oberski, D.** (2013). Issues in Structural Equation Modeling of Complex Survey Data. In Proceedings of the 59th World Congress of Statistics.

- **Bosworth, B.** and **Burke, K.** (2012). Changing Sources of Income in the Elderly Population. Center for Retirement Research at Boston College *working paper*, (2012-27).
- Brown, S.L. (2010). Marriage and Child Well-Being: Research and Policy Perspectives. *Journal of Marriage* and Family, 72(5): 1059-1077.
- **Bruce D. Meyer** and **James X. Sullivan** (2003). "Measuring the Well-Being of the Poor using Income and Consumption". Conference on Income Volatility and its Implications for Food Assistance, 2-3 May 2002 in Washington, DC.
- Buhmann, B., Rainwater, L., Schmaus, G. and T. Smeeding (1988), "Equivalence Scales, Well-Being, Inequality and Poverty: Sensitivity Across Ten Countries Using the Luxembourg Income Study (LIS) Database", *Review of Income and Wealth*, **94**: 115-142.
- **Byrne, B.M.** (2010). Structural Equation Modelling with AMOS: Basic Concepts, Applications and Programming, 2nd edition, Routledge, New York.
- C. Lwechungura Kamuzora and William Mkanta. (2000). Poverty and Household/Family Size in Tanzania: Multiple Responses to Population Pressure? Research On Poverty Alleviation. *Research Report No.* 00.4
- Cheung, F. and Lucas, R. E. (2016). Income Inequality is associated with Stronger Social Comparison Effects: The Effect of Relative Income on Life Satisfaction. *Journal of Personality and Social Psychology*, **110**(2): 332.
- **Copestake, J.** (2007). What is the relevance of well-being to microfinance? ESC Research Group on Well-being in Developing Countries. University of Bath. UK.
- Coulter, F.A.E., Cowell, F.A. and S.P. Jenkins (1992). "Equivalence Scale Relativities and the Extent of Inequality and Poverty", *Economic Journal*, **102**: 1067-1082.
- **Chant, Sylva H.** (2003). Female Household Headship and The Feminisation of Poverty: Facts, Fictions and Forwards Strategies [Online] London: LSE Research Online.
- Datt, G. and Jolliffe, D. (1999). Determinants of Poverty in Egypt:1997. NDRF *Working Paper No.* 75, October.
- **Daud, A.** (2004). Factors that motivate women entrepreneurs in the market of Tanah Merah, Kelantan. Universiti Teknologi Mara.
- **De Neubourg, C., M. De Milliano,** et al. (2014). Lost Dimensions: Consolidating progress in multidimensional poverty research. Innocenti *Working Paper No.* 2014-04. Florence, UNICEF Office of Research.

- Diener, E. and Biswas-Diener, R. (2002). Does money increase subjective well-being? *Social indicators research*, 57(2): 119-169.
- **Dorosh, P.A.** and **Lundberg, M.K.** (1996). More than just Peanuts (Groundnuts: Aid Flows and Policy Reform in The Gambia. Economic Reform and the Poor in Africa, Clarendon Press: Oxford.
- Dreze, J. and P.V. Srinivasan (1997). "Widowhood and Poverty in Rural India: Some Inferences from Household Survey Data", *Journal of Development Economics*, 54(2): 217-234.
- **Dunga, S.H.** (2017). A Gender and Marital Status Analysis of Household Income in a Low-Income Township. Studia Universitatis Babes-Bolyai Oeconomica, **62**(1): 20-30.
- Eko Suprayitno, Mohamed Aslam and Azhar Harun. (2017). Zakāh and SDGs: Zakāh Impact on Human Development in the Five States of Malaysia. *International Journal of Zakāh* **2**(**1**): 61-69.
- Eko Suprayitno, Radiah Abdul Kader and Azhar Harun. (2013). The Impact of Zakāh on Aggregate Consumption In Malaysia. *Journal of Islamic economics, banking and finance*, 9(1): Jan-Mar 2013.
- Elder Jr, G.H., Conger, R.D., Foster, E.M. and Ardelt, M. (1992). Families under Economic Pressure. *Journal* of Family Issues, 13(1): 5-37.
- Emini, C. and Fofack, H. (2004). A Social Financial Accounting Matrix for the Integrated Macroeconomic Model of Poverty Analysis: Application to Cameroon with a Fixed Price Multiplier Analysis. World Bank Policy Research. *Working Paper*, 3219.
- Endut, N., Azmawati, A.A. and Hashim, I.H.M. (2015). Formal and Informal Support Systems for Single Women and Single Mothers in Malaysia. In SHS Web of Conferences (Vol. 18, pp. 03002). EDP Sciences.
- Erola, J., Jalonen, S. and Lehti, H. (2016). Parental education, class and income at the beginning of the life course and children's success. *Research in Social Stratification and Mobility*, 44: 33-43.
- Etikan, I. and Babtope, O. (2019). A basic approach in sampling methodology and sample size calculation. *Med Life Clin*, **1**(2): 1006
- Fan, S., Hazell, P. and Thorat, S. K. (2000). Impact of public expenditure on poverty in rural India. *Economic* and Political Weekly, 3581-3588.
- Field, A. (2013). Discovering statistics with IBM SPSS Statistics. SAGE.

- Fuadah Johari, Ahmad Fahme Mohd Ali, Muhammad Ridhwan Ab. Aziz (2015). "The Role Of Zakāh Distribution Among Muallaf (New Convert) In Reducing Poverty In Selangor, Malaysia". İktisat Politikası Araştırmaları Dergisi (*Journal of Economic Policy Research*). 2(3): 39-56.
- Fuwa, N. (2000). "Poverty and Heterogeneity Among Women Household Heads Revisited: The Case of Panama", World Development, 28(8): 1515-1542.
- Gang, I.N., Sen, K. and Yun, M-S (2004), Caste, Ethnicity and Poverty in Rural India. (See: www.wm.edu/ economics/seminar/papers/gang.pdf)
- Geda, A., de Jong, N., Kimenyi, M. S. and Mwabu, G. (2005). Determinants of poverty in Kenya: A household level analysis.
- Guttman, L. and Levy, S. (1982). On the Definition and Varieties of Attitude and Well-being. *Social Indicators Research*, 10: 159-174.
- Hahs-Vaughn, D. and Lomax, R. (2006). Utilization of Sample Weights in Single-Level Structural Equation Modeling. *The Journal of Experimental Education*, 74(2): 161-190.
- Hair, J.F., Hult, G.T., Ringle, C. and Sarstedt, M. (2014). A primer on partial least squares structural equation modelling (PLS-SEM). Thousand Oaks: Sage.
- Hair, J.F., Ringle, C. M. and Sarstedt, M. (2011). PLS-SEM: Indeed a silver bullet. *Journal of Marketing Theory and Practice*, **19**(2): 139-151. http://dx.doi.org/ 10.2753/MTP1069- 6679190202.
- Hair, J.F., Sarstedt, M., Ringle, C. and Mena, J.A. (2012). An evaluation of partial least squares structural equation modelling in marketing research. *Journal of the Academy of Marketing Science*, **40**(3): 414-433.
- Hair, J.F., William, B.C., Barry, B.J. and Rolph, A.E. (2010). *Multivariate Data Analysis* (7th ed.). Pearson Prentice Hall.
- Hairunnizam Wahid, Radiah Abdul Kader and Sanep Ahmad.(2012). Location of Zakāh distribution, religiosity, quality of life and attitude change. (Perceptions of Zakāh recipients in Malaysia). The 13th Malaysia-Indonesia Conference on Economics, Management and Accounting (MIICEMA) 2012.
- Hassan, M.K. (2010). An integrated model of poverty reduction combining Zakāh, Awqaf, and Micro-Finance. Seven International Conferences-Tawhidi Zakāh Epistemology and Waqaf Economics, Bangladesh. http://www.kep.org.sa/en/SideMenuContent_en.aspx? MenuID=97&MenuTitle=Education%20in%20Islam.

- Hafizah Zainal, Azizi Abu Bakar, Ram Al Jaffri Saad. (2016). Reputation, satisfaction of zakāh distribution, and service quality as determinants of stakeholders' trust in zakāh institutions. *International journal of economic and financial issues*, 2016, 6(S7): 72-76.
- Henseler, J. and Sarstedt, M. (2013). Goodness-of-fit indices for partial least squares path modelling. Computational Statistics, 28(2): 565-580.
- Henseler, J., Hubona, G. and Ray, P.A. (2016). Using PLS path modelling in new technology research: updated guidelines. Industrial Management & Data Systems, 116(1), 2-20. http://dx.doi.org/10.1108/IMDS -09-2015-0382.
- Henseler, J., Ringle, C.M. and Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modelling. *Journal* of the Academy of Marketing Science, 43(1): 115-135.
- Hidrobo, M., Peterman, A. and Heise, L. (2016). The effect of money, vouchers, and food transfers on intimate partner violence: evidence from a randomized experiment in northern Ecuador. *American Economic Journal: Applied Economics*, **8**(3): 284-303.
- Holden KC, Smock PJ. (1991). The economic costs of marital dissolution: Why do women bear a disproportionate cost? Annu Rev Sociol. 1991, 17: 51-78.
- **Ibrahim, P.** (2007). Peranan agihan zakāh dalam meningkatkan tahap ekonomi ummah. Paper presented at Konvensyen Zakāh dan Cukai Peringkat Kebangsaan 2007, Kuala Lumpur, Malaysia, 2007.
- **Iceland, J.** (2013). Poverty in America: A Handbook. Univ. of California Press.
- **Iqbal, Munawar** (1986). Distributive Justice and Need Satisfaction in an Islamic Economy. Leicester: The Islamic Foundation.
- Ishida, A., Law, S.H. and Aita, Y. (2003). Changes in food consumption expenditure in Malaysia. Agribusiness: *An International Journal*, **19**(1): 61-76.
- Iskandar Hasan Tan Abdullah, Nur Azuki Yusoff, Rosdi Abd Rahman. (2016). Quality of life and income level of poverty in Kelantan State, Malaysia. *Journal of applied environmental and biological sciences. J. Appl. Environ. Biol. Sci*, 2016, 6(6S): 51-57. ISSN: 2090-4274.
- Joana Costa and Marcelo Medeiros, (2008). Is there a feminization of poverty in Latin America? *World Development* **36** (1): 115{127.
- Johari, F., Ab Aziz, M.R. and Ali, A. F.M. (2014). The role of zakāh in reducing poverty and income inequality among new converts (muallaf) in Selangor, *Malaysia*. *Online Journal of Islamic Studies Research*, 1(3): 43-56.

- Johari, Fuadah (2004) Keberkesanan zakāh dalam mengatas masalah kemiskinan di negeri Melaka. Master's thesis, University of Malaya. Download from http://student srepo.um.edu.my/1839
- Jones SJ, Frongillo EA. (2006). The modifying effects of food stamp program participation on the relationship between food insecurity and weight change in women. J Nutr. 2006, **136**:1091-4.
- Kamuzora, C.L. and S. Gwalema (1998). Aggravation of Poverty in Rural Bukoba District, Tanzania: Labour Constraints, Population Dynamics and the AIDS Epidemic. *Research Report*, 22 June 1998. (Dar es Salaam: REPOA).
- Kasy, M. and Ramos-Chaves, A. (2014). The Impact of the Changing Family Structures on Income Distribution among Costa Rican Women 1993-2009. *Feminist Economics*, 20(2): 122-144.
- Kock, N. (2015). Standard method bias in PLS-SEM: an approach to comprehensive collinearity assessment. *International Journal of e-Collaboration*, **11**(**4**): 1-10. http://dx.doi.org/10. 4018/ijec.2015100101.
- Korpi, W., Ferrarini, T. and Englund, S. (2013). Women's opportunities under different constellations of family policies: gender, class, and inequality trade-offs in western countries re-examined. *Social Politics: International Studies in Gender, State & Society*, **20**(1): 1-40.
- Kjellgren, A., C. Jones-Pauly, H. El-Tayeb Alyn, E. Tadesse, and A. Vermehren. (2014). "Evaluation of Sudan's Social Safety Net." *Social Protection and Labor Working Paper No. 1415.* Washington, DC: World Bank.
- Krejcie, R.V. and Morgan, D.W. (1970). "Sample size determination for research activities". *Educational and Psychological Measurement*, 1-4.
- Laabas, B. and Limam, I. (2004). Impact of Public Policies on Poverty, Income Distribution and Growth. Paper prepared for the IFPRI/API collaborative research project: "Public Policy and Poverty Reduction in the Arab Region". Arab Planning Institute. Kuwait.
- Lancaster, G., Ray, R. and R. Valenzuela (1999). "A Cross Country Study of Household Poverty and Inequality on Unit Record Household Budget Data", *Economic Development and Cultural Change*, **48**(1): 177-208.
- Lanjouw, P. and M. Ravallion (1995). "Poverty and Household Size", *Economic Journal*, 105: 1415-1434.
- Lee, C. and Hallak, R. (2018). Investigating the moderating role of education on a structural model of restaurant performance using multigroup PLS-SEM analysis. *Journal of Business Research*, 88: 298-305.

- Levin, J. and Bigsten, A. (2000). Growth, income distribution, and poverty: a review. report nr: *Working Papers in Economics*, (32).
- Light A. (2004). Gender differences in the income premium of marriage and cohabitation. *Demography*. 41: 263-84.
- Lowry, P. B. and Gaskin, J. (2014). Partial least squares (PLS) structural equation modelling (SEM) for building and testing causal behavioral theory: When to choose it and how to use it. *IEEE Transactions on Professional Communication*, 57(2): 123-146.
- **M. Akram, M.** and **Afzal, M.** (2014). "Dynamic role of Zakāt in poverty alleviation: A Case Study of Pakistan", University Library of Munich, Germany.
- Machado, Anna Carolina; Bilo, Charlotte; Helmy, Imane (2018). The role of *zakāh* in social protection provision: A comparison between Jordan, Palestine and Sudan, *Working Paper*, No. 168, International Policy Centre for Inclusive Growth (IPC-IG), Brasilia.
- Malapit, H.J.L., Kadiyala, S., Quisumbing, A.R., Cunningham, K. and Tyagi, P. (2015). Women's empowerment mitigates the adverse effects of low production diversity on maternal and child nutrition in Nepal. *The Journal of Development Studies*, 51(8): 1097-1123.
- Manning, W. and Brown, S. (2003). The Economic wellbeing of children in cohabiting families. Center for Family and Demographic Research, *Working Papers*, (03-05).
- Mariano Rojas (2004). "The Complexity of Well-being: A Life Satisfaction Conception and a Domains-of-Life Approach". Paper for the International Workshop on Well-being Research in Developing Countries.
- Meenakshi, J.V. and Ray, R. (2002). Impact of household size and composition on poverty in rural India. *Journal* of Policy Modeling, 24(6): 539-559.
- Mentzakis, E. and Moro, M. (2009). The poor, the rich, and the happy: Exploring the link between income and subjective well-being. The Journal of Socio-Economics, 38(1): 147-58.doi:10.1016/j.socec.2008.07.010
- Mergenhagen, P.M., Lee, B.A., Gove, W.R. (1985). Till death do us part: Recent changes in the relationship between marital status and mortality. Sociology and Social Research, **70**: 53-56.
- Meyer, B. and Sullivan, J. (2003). Measuring the Well-Being of the Poor Using Income and Consumption. *The Journal of Human Resources*, **38**: 1180-1220. doi:10.2307/3558985.

- Mistry, R.S., Vandewater, E.A., Huston, A.C. and McLoyd, V.C. (2002). Children's economic well-being and social adjustment: The role of family process in an ethnically diverse low-income sample. *Child Development*, 73(3): 935-951.
- Mohd, S. (2014). Poverty issues among Malaysian elderly. Proceeding of the Social Sciences Research ICSSR 2014 (e-ISBN 978-967-11768-7-0). 9-10 June 2014, Kota Kinabalu, Sabah, MALAYSIA. Organized by http://WorldConferences.net.
- **Mok, T.Y.** (2009). Poverty Lines, Household Economies of Scale and Urban Poverty in Malaysia. (Doctoral dissertation, Lincoln University, 2009).
- Mohd Amar Aziz, Badariah Haji Din, Kamaruddin Abdulsomad. (2018). "The contribution of location management system on the performance of zakāh institutions". *International Journal of Supply Chain Management*, **7**(1): February 2018.
- Muhammad Quraisy, Syed Othman Alhabshi, Shaikh Hamzah Abdul Razak. (2017). The impact of Islamic microfinance in improving well-being and quality of life: Case Study of Islamic Financial Cooperative (BMT) In Indonesia. Southeast Asia Journal of Contemporary Business, *Economics and Law*, vol. 13, issue 3 (August).
- Muhammad, Aliyu Dahir. Amin, M.D. Fuoad. Oladukun, Nafiu (2013). 'Zakāh distribution among the Asnaf of Wilayah and Selangor: Analytical Hierarchy Process (AHP)'.
- Mukaramah, H., Zafarullah, A.A. and Nor'Aznin, A.B. (2011). Impact of household income distribution on government expenditure by component in Malaysia. *International Review of Business Research*, **7**(4): 140-165.
- Nizam, J.M., Amirul, A., Hardi, A., Nurul, S.C. (2011), A study on the factors attribute to non-participation of zakāh income among the Muslim Community in Selangor. 2nd International Conference on Business and Economic Research (ICBER), Proceeding. pp. 450-461.
- **OECD** (2013). "Economic well-being", in OECD Framework for Household Income, Consumption and Wealth Distribution Statistics, OECD Publishing, Paris.
- **Ogbeibu, S., Senadjki, A.** and **Gaskin, J.** (2018). The moderating effect of benevolence on the impact of organizational culture on employee creativity. *Journal of Business Research*, **90**: 334-346.
- Oktaviani, R., Hakim, D.B. and Siregar, H. Sahara (2004). The Impact of Reducing Oil Subsidy on Indonesian Macroeconomic Performance, Agricultural Sector and Poverty Incidences (A Recursive Dynamic Computable General Equilibrium Analysis). oktavia@ indo.net.id.

- Pahl J. (1995). Her money, His Money: Recent research on financial organization in marriage. J Econ Psychol 1995; 16(3): 361-76.
- **Pallant, J.** (2013). SPSS Survival Manual. McGraw-Hill Education (UK).
- Pamela J. Smock, Wendy D. Manning and Sanjiv Gupta (1999). American Sociological Review, 64(6): (Dec., 1999): 794-812.
- Patmawati Hj Ibrahim (2008). Pembangunan Ekonomi Melalui Agihan Zakāh: *Tinjauan Empirikal. Shariah Journal*, 2008. 16(2): 223-244-223.
- **Paswan, A.** (2009). Confirmatory Factor Analysis and Structural Equations Modeling, An Introduction, Department of Marketing and Logistics, COB, University of North Texas, USA.
- **Pelling, M.** and **Wisner, B.** (2012). Disaster risk reduction: Cases from urban Africa. Routledge.
- Podsakoff, P.M., MacKenzie, S.B., Lee, J.Y. and Podsakoff, N.P. (2003). Common method biases in behavioural research: A critical review of the literature and recommended remedies. *Journal of Applied Psychology*, 88(5): 879-903. http://dx.doi.org/10.1037/ 0021-9010.88.5.879.
- Ragayah Haji Mat Zin. (2007). Understanding the formulation of the revised poverty line in Malaysia. Akademika 70 (Januari) 2007: 21-39
- Ravallion, M. and Benu Bidani (1994). How Robust is a Poverty Profile? *World Bank Economic Review* 8(75-102).
- Raedah, S., Noormala, A., Marziana, M. (2011). A Study on Zakāh of Employment Income: Factors influencing the intention of scholars to pay Zakāh. 2nd International Conference on Business and Economic Research, (2nd ICBER, 2011), Proceeding. pp. 2492-2507.
- Rodriguez, J.G. (2002). The Determinants of Poverty in Mexico.www.gdnet.org/pdf/2002AwardsWinners/Gro wthInequalityPoverty/Jorge_garza_rodriguez_paper.pdf.
- Rokhman, W. (2014). The Effect of Islamic Microfinance on Poverty Alleviation: Study in Indonesia, *International Journal of Economics and Business*, Volume 11, Issue 2.
- Roloff, O. (2002) Mehr als dreissig Jahre Fiskalpolitik in Deutschland. Die unerfüllten Versprechungen der Dogmatiker, Applied Economics Quarterly, 48(2): 147-168.
- **Romulo A. Virola** and **Arturo M. Martinez Jr.** (2007). "Population and Poverty Nexus: Does Family Size Matter?" Paper presented at the 10th National Convention on Statistics at the EDSA Shangri-la Plaza Hotel, Mandaluyong City, Philippines, October 1-2, 2007.
- Saad, R.A. and Haniffa, R. (2014). Determinants of zakāh (Islamic tax) compliance behavior. Journal of Islam, accounting and business. 5(2): 182-193.

- Sani Adamu Muhammad and Ram Al-Jaffri Saad. (2016). Moderating effect of attitude towards zakāh payment on the relationship between moral reasoning and intention to pay *zakāh*. 3rd Global Conference on Business and Social Science-2015, GCBSS-2015, 16-17 December 2015, Kuala Lumpur, Malaysia.
- Salman Ahmed Shaikh, and Abdul Ghafar Ismail. (2017) Role of *zakāh* in sustainable development goals. *International Journal of Zakāh*. 2017, **2**(2): 1-9.
- Sarstedt, M., Ringle, C.M., Smith, D., Reams, R. and Hair, J.F. (2014). Partial least squares structural equation modelling (PLS-SEM): A useful tool for family business researchers. *Journal of Family Business Strategy*, 5: 105-115. http://dx.doi.org/10.1016/ j.jfbs.2014.01.002.
- Sohag, K., Mahmud, K.T., Alam, F. and Samargandi, N. (2015). Can the *zakāh* system alleviate rural poverty in Bangladesh? A propensity score matching approach. *Journal of Poverty*, **19**(**3**): 261-277.
- Shaikh, S. A. (2016). "Collectable zakāh in OIC countries for poverty alleviation: A Primer on Empirical Estimation". *International Journal of Zakāt*, 1(1): 17-35.
- Sheriff Muhammad Ibrahim (2015). The role of Zakāh in establishing social well-being and economic sustainability. International Journal of Management and Commerce Innovations ISSN 2348-7585 (Online) Vol. 3, Issue 1: 437-441.
- Shirazi, N.S. (2006). Reducing poverty through the institution of Zakāh. Islamabad International Institute of Islamic Economics and Birmingham. Religions and Development Research Program (mimeo).
- Shireen Safdar, Nisar Ahmad and Falak Sher. (2012). Estimation of urban and rural expenditure and size elasticities of food items in Pakistan: Evidence from Pslm Data. *Academic Research International.* **3**(2): September 2012.
- **Siddiqui, M.N.** (1978). Some Aspects of the Islamic Economy, Second Edition, Lahore, Pakistan: Islamic Publications Ltd.
- Suhaimi Abd Samad, Norma Mansor (2013). Population aging and social protection in Malaysia. Malaysian *Journal of Economic Studies* 50(2): 139-156.
- Suma Devi S, Aaryaambhaa C (2015). A study on income sources of the elderly in Coimbatore city. *International journal of applied research 2016*; 2(1): 175-177.
- Sun, W., Li, D., Zhang, W., Bao, Z. and Wang, Y. (2015). Family material difficulties and problem behaviors of Chinese adolescents: A moderated mediation analysis. *PloS one*, **10**(5): e0128024.

- Suprayitno, E., Kader, R. A. and Harun, A. (2013). The impact of *zakāh* on aggregate consumption in Malaysia. *Journal of Islamic Economics, Banking and Finance*, 9(1): 1-24.
- Suprayitno, E., Aslam, M. and Harun, A. (2017). Zakāh and SDGs: zakāh impact on human development in the five states of Malaysia. *International Journal of Zakāh*, 2(1): 61-69.
- Taylor, M.P., Jenkins, S.P. and Sacker, A. (2011). Financial capability and psychological health. *Journal* of economic psychology, 32(5): 710-723.
- Tiwari, S., Daidone, S., Ruvalcaba, M. A., Prifti, E., Handa, S., Davis, B., ... and Seidenfeld, D. (2016). Impact of cash transfer programs on food security and nutrition in sub-Saharan Africa: A cross-country analysis. *Global Food Security*, 11: 72-83.
- **Turkawi, A.** (2015). Social protection and safety nets in Sudan. Brighton: Institute of Development Studies.
- **Ullman, J.B.** (2006). *Structural equation modeling: Demystified and applied, Journal of Personality Assessment.*
- Usman, A.S. and Tasmin, R. (2016). The role of Islamic microfinance in enhancing human development in Muslim countries. *Journal of Islamic Finance*, 2016, 5(1): 053-062. IIUM Institute of Islamic Banking and Finance, Malaysia.
- Utsey, S.O. and Constantine, M.G. (2008). Mediating and moderating effects of racism-related stress on the relationship between poverty-related risk factors and subjective well-being in a community sample of African Americans. *Journal of Loss and Trauma*, 13(2-3): 186-204.
- Waite LJ. (1995). Does marriage matter? Demography. 1995; 32:483-507.
- Wahab, N.A. and Rahman, A.R.A. (2011). A framework for analyzing the effectiveness and governance of zakāh institutions. *Journal of Islamic Accounting and Business Research*.
- Whelan, C.T., B. Nolan, et al. (2014). "Measuring multidimensional poverty in Europe: an application of the adjusted enumeration approach". *Journal of European Social Policy*, 24(2): 183-197.
- Wight, V., Kaushal, N., Waldfogel, J. and Garfinkel, I. (2014). Understanding the link between poverty and food insecurity among children: Does the definition of poverty matter? *Journal of Children and Poverty*, 20(1): 1-20.
- Yong, G.A. and Pearce, S. (2013). A beginner's guide to factor analysis: focusing on exploratory factor analysis. *Tutorial in Quantitative Methods for Psychology*, 9(2): 79-94.

- Yusoff, M.B. (2006). Tax policy in an Islamic economy and the role of *zakāh*. *International Journal of Economics*, *Management and Accounting*, 14(2).
- Yusoff, M. B. (2010). An Analysis of Zakót Expenditure And Real Output: Theory And Evidence. *International Journal of Economics, Management and Accounting*, 18(2).
- Yusoff, M. B. (2011). Zakāh spending, school enrollment and economic growth in Malaysia. International Journal of Business and Social Science, 2(6): 175-181.
- Zagorsky JL. (2005). The impact of marriage and divorce on wealth. J Sociol (Melb). 2005; 41:406-24.

- Zainol, B., Kamil, M. and Fardahwati, S. (2009). Predicting compliance intention on *zakāh* on employment income in Malaysia: and application of reasoned action theory. *Jurnal Pengurusan*, 28(1): 85-102.
- Zainol, B. and Faridahwati, M.S. (2013). Using the theory of reasoned action to explain taxpayers' intention to comply with the goods and services tax (GST). *Middle East Journal of Scientific Research*, 17(3): 387-394.
- Zikmund, W.G., Babin, B.J., Carr, J.C. and Griffin, M. (2013). *Research methods in business* (ninth edition). Erin Joyner, Cangage Learning.

Ahmad Fahme Mohd Ali, PhD, is a Senior Lecturer at Faculty of Hospitality, Tourism and Wellness, University Malaysia Kelantan (UMK). He received his first degree in Bachelor of Business Administration (International Business) from Southern Cross University, Australia in 2006 and MSc (Economic) in 2011 from Universiti Putra Malaysia, specializing in Development Economic. In 2016, he was awarded a doctoral degree in Development Economic, from Universiti Sains Islam Malaysia (USIM). His area of expertise is in the field of Poverty and Islamic Economic Development specifically on zakat distribution and zakat poverty line (Haddul Kifayah). He was previously a Post- Doctoral at Institute for Poverty Research and Management (InsPeK), UMK. He has also presented and published several articles in chapter in books and articles in various national and international conferences and journals. He can be reached at fahme.ma@umk.edu.my.

W.A. Amir Zal, PhD, is an Associate Professor in the community development discipline and a university lecturer in the Faculty of Language Studies and Human Development, Universiti Malaysia Kelantan, Malaysia. Currently, he is the director of the Institute for Poverty Research and Management (InsPeK) Universiti Malaysia Kelantan and Secretary-General of the Regional Network on Poverty Eradication (RENPER). His research and publication primarily emphasise exclusion and inclusion, poverty, empowerment, and development plans for vulnerable communities. He is also actively involved in developing various policies to empower a community across the issues. He can be contacted through waamirzal@umk.edu.my or waamirzal@gmail.com.

Nurhanan Abdul Rahman, PhD, currently working at Universiti Malaysia Kelantan (UMK) as a senior lecturer. She obtained BSc majoring in Biology (2002) and pursued with Masters's study in Microbiology (2006) from Universiti Sains Malaysia (USM). She temporarily worked as a research assistant at the School of Health Sciences, Universiti Sains Malaysia, before pursuing her Ph.D. with specialization in Nutritional Biochemistry and being conferred the doctorate in 2015 from the same institute. Her research interests focus on utilization of plant and their by-products in food products development, particularly in improving nutritional, physical and microbiological qualities of the food products. As a senior lecturer at Faculty of Agro-Based Industry, UMK, she delivers lectures on topics related to food science and nutrition. Recently, she starts to explore community research and involved in several knowledge transfer programs to implement and share her knowledge in foods and nutrition with the poor communities residing in Kelantan state. She can be contacted through nurhanan.ar@ umk.edu.my.

Tengku Fauzan Tengku Anuar, PhD, is the Head of Research Division, Institute of Research and Poverty Management (InsPeK). She also serves as a Senior Lecturer at the Faculty of Creative Technology & Heritage, Universiti Malaysia Kelantan (UMK). She obtained Bachelor from University Malaysia Sabah and pursued with Masters's study in Visual Communication and New Media from Universiti Teknologi Mara (Malaysia). She obtained her PhD from Dongseo University, Korea in Visual Communication and New Media. She has a 12 year of experience in Visual Communication and New Media. She has a 12 year of experience in Visual Communication and New Media. She started engaging with InsPeK in 2018 as a researcher and until now and the main activities she focuses are involving community development. She is passionate about serving the community through capacity building and development programs. She can be contacted through tengkufauzan@umk.edu.my.

Hafizi bin Mat Salleh, is a Social Research Officer at the Institute for Poverty Research and Management (InsPeK), Universiti Malaysia Kelantan (UMK). He obtained his Bachelor's Degree and Master in Geography from Universiti Kebangsaan Malaysia. He mostly involved in developing the skills and capacity of the poor community programs through training activities and poverty mapping under InsPeK. He can be contacted through hafizi.ms@umk.edu.my.

Siti Asma binti Md Rasdi, is a Social Research Officer at the Institute for Poverty Research and Management (InsPeK), Universiti Malaysia Kelantan (UMK). She obtained her Bachelor's Degree in Information Technology and she mostly involved in building the capacity of the poor community through training activities and poverty mapping under InsPeK. She also responsible for managing the poor database which was built by InsPeK for building capacity programs. She can be contacted through asma.mr@umk.edu.my.

التأثير المعتدل لتوزيع الزكاة على الرفاهية الاقتصادية للفقراء: تحليل من ولاية كلانتان بماليزيا

أحمد فهمي محمد علي⁽¹⁾؛ أمير زال⁽¹⁾؛ نورحنان حامد^(ب)؛ تنكو فوزان أنور^(ب)؛ حفيظي صالح^(ج) وستي أسماء رشدي^(ج) ^(m) أستاذ مساعد، كلية الضيافة؛ ^(m) أستاذ مشارك، مركز الدراسات اللغوية ^(ب) أستاذ مساعد، كلية الدراسات الزراعية؛ ^(ب) أستاذ مساعد، كلية تقنية الإبداع والتراث ^(ج) أستاذ مساعد، معهد أبحاث وإدارة الفقر؛ ^(ج) أستاذ مساعد، معهد أبحاث وإدارة الفقر ^(ج) أستاذ مساعد، معهد أبحاث وإدارة الفقر؛ ^(ج) أستاذ مساعد، معهد أبحاث وإدارة الفقر

المستخلص. هَدَفت الورقة إلى دراسة التأثير الوسيط لتوزيع الزكاة على بُنية الأسرة والرفاهية الاقتصادية للفقراء في ولاية كلانتان بماليزيا. تم اختيار ولاية كلانتان للمسح نظرًا لوجود أعلى معدلات للفقر فيها أكثر من ولايات ماليزيا الأخرى. اعتمد تحليل الدراسة على تقنية نمذجة المعادلات الهيكلية القائمة على التباين (VB-SEM)، بينما أجريت عملية استخراج البيانات والإحصاءات الوصفية، والبيانات الديموغرافية لهذه الدراسة أجريت عملية استخراج البيانات والإحصاءات الوصفية، والبيانات الديموغرافية لهذه الدراسة أن أبعاد القائمة على التباين (VB-SEM)، بينما أجريت عملية استخراج البيانات والإحصاءات الوصفية، والبيانات الديموغرافية لهذه الدراسة باستخدام برنامج (SPSS). أَظُهرت نتائج الدراسة أن أبعاد المساعدة الزكوية، وهيكل الأسرة تؤثران تأثيرًا إيجابيًا ومعنويًا على الرفاهية الاقتصادية للأسر الفقيرة. كما تُشير النتائج كذلك إلى أن التأثير المباشر للعمر، والحالة الاجتماعية، والجنس ليس مهمًا بسبب عدم كفاية الرفاه الاقتصادي الذي لا يلبي احتياجاتهم الأساسية. في حين يُشير الدور الوسيط للزكاة إلى أن التأثير غير مبام المباسية. في حين يُشير الدور الوسيط للزكاة المباسية المباسية. في حين يُشير النتائج كذلك إلى أن التأثير المباشر للعمر، والحالة الاجتماعية، والجنس ليس مهمًا بسبب عدم كفاية الرفاه الاقتصادي الذي لا يلبي احتياجاتهم الأساسية. في حين يُشير الدور الوسيط للزكاة إلى أن العمر والحالة الاجتماعية، والجنس ليم مهمًا بسبب عدم كفاية الرفاه الاقتصادي الذي لا يلبي احتياجاتهم الأساسية. في حين يُشير الدور الوسيط للزكاة إلى أن العمر والحالة الاجتماعية، والجنس لهما تأثير غير مباشر ومهم، وتأثير مباشر غير مهم على الرفاهية الاقتصادي أن التأثير غير المباشر لبنية الأسرة على الرفاه الاقتصادي تقوم الزكاة الرفاهية القتصادية، ما يعني أن التأثير غير المباشر لبنية الأسرة على الرفاه الاقتصادي تقوم الزكاة الرفاهية الاقتصادية، مما يعني أن التأثير غير المباشرة على الرفاه الاقتصادي تقوم الزكاة الرفاهية الاقتصادية، مما يعني أن التأثير غير المباشر لبنية الأسرة على الرفاه الاقتصادي تقوم الزكاة الرفاهية الرفاه الاقتصادي التأثير في ألماسية الأسرة ماي الرفاه الاقتصادي الي ألماسية الأسرة على الرفاه الاقتصادي الركان المثان ما لي ألفي ألماسية الأسرة على الرفاه الاقتصادي الكان الماش

الكلمات الدَّلة: نموذج المعادلة الهيكلية (SEM)، الفقير والمسكين، توزيع الزكاة، الفقر، كلانتان، ماليزيا.

تصنيف D13, D31, I32 :JEL

تصنيف E12, E15, N4 :KAUJIE